Author's response to reviews

Title: Capacity mapping of national ethics committees in the Eastern Mediterranean Region

Authors:

Alaa Abou-Zeid (alaabouzeid@yahoo.com)
Mohammad Afzal (AFZALMO@emro.who.int)
Henry Silverman (hsilverm@medicine.umaryland.edu)

Version: 3 Date: 26 June 2009

Author's response to reviews:

We have made the following changes:

Author list: removed job titles
Authors' contributions: we have indicated that all authors read and approved the final manuscript.
We have answered the queries and made the appropriate changes to the references.
The Tables are now included in the body of the manuscript.

Title is in proper format

Key words eliminated
email addresses are now provided.
All underlining removed
We have performed a final editing of the manuscript.